Author's response to reviews

Title: Feasibility of provision of sustainable intra-dermal post exposure prophylaxis against rabies at primary care level -evidence from rural Haryana, India

Authors:

Harshal Salve Dr (harshalsalve@ymail.com)
Sanjeev Kumar Dr (docsanjiv@gmail.com)
Rizwan SA Dr (sarizwan1986@gmail.com)
Sanjay K Rai Dr (drsanjay.aiims@gmail.com)
Shashi Kant Dr (skant76@gmail.com)
Chandrakant S Pandav Dr (cpandav@iqplusin.org)

Version: 3 Date: 16 October 2013

Author's response to reviews:

Covering letter
To
The Editor
BMC Health Service Research
Dear Sir,

We intend to publish an article entitled “Feasibility of provision of sustainable intra-dermal post exposure prophylaxis against rabies at primary care level –evidence from rural Haryana, India” in your esteemed journal as research article.

This study documents the process of implementation cost effective and sustainable health service of intra-dermal anti rabies pre-exposure prophylaxis at primary care level. This is an important health service related study in the light of high burden of unmet need of animal bite management at community level in India. This study provided initial evidence of cost-effective and sustainable model of health service delivery of intra-dermal anti rabies pre-exposure prophylaxis at primary care in India which will help policy makers for addressing rabies problem in India.

We hereby transfer, assign, or otherwise convey all copyright ownership, including any and all rights incidental thereto, exclusively to the journal, in the event that such work is published by the journal.

Looking forward to hearing from you soon
Thanking you,

With regards

Dr. Harshal Salve